Author’s response to reviews

Title: Severe thoracic trauma caused left pneumonectomy complicated by right traumatic wet lung, reversed by extracorporeal membrane oxygenation support—a case report

Authors:

Feng Yun Wang (dr-w@qq.com)
Bin Fang (dr_fangbin@163.com)
Zhi Hui Yu (dr_zhihuiyu@163.com)
Jing Song Shao (dr_shaojingsong@163.com)
Wei Biao Wen (wwbiao@fsyy.com)
Li Xin Zhou (drzlxin@126.com)

Version: 3 Date: 16 Jan 2019

Author’s response to reviews:

Dear Editor,

Thank you very much for your help.

First, the manuscript was revised according to the reviewers' advice and the language was edited carefully.

Second, the figures were uploaded separately as required. 'Figure Legends' were written after the references.

Third, we did not intend to remove author Wen Wei Biao, his name did not list on the first paper was just our carelessness during revision.

For that, we are truly sorry. Haste makes waste! We'll do better in the future.

Thank you again for your help and consideration!